Author's response to reviews

Title: Assessement of postoperative long-term survival quality and complications associated with radical antegrade modular pancreatosplenectomy and distal pancreatectomy: a Metameta-analysis and systematic review

Authors:

Zhou Quanyu (1652540230@qq.com)
Lei Zehua (leitsehua@126.com)
Gong Jie (827361656@qq.com)
Xie Qingyun (410321477@qq.com)
Liu Yu (243171791@qq.com)
Wang Qing (309696055@QQ.COM)
Zhao Xin (845489808@qq.com)

Version: 2 Date: 15 Aug 2018

Author's response to reviews:

Dear reviewer:

I am very grateful to your comments for the manuscript. According with your advice, we amended the relevant part in manuscript. Some of your questions were answered below.

1. Response to comment: the current submission contains 34% textual overlap with other previously published works.

Reply: We carefully resummarize the article language, especially the methods section.

2. Response to comment: please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

Reply: The email addresses of all authors have been indicated on the title page.
3. Response to comment: Author LY is missing from the list of author contributions.

Reply: We carefully reviewed each author's research contributions and apologized for our omissions.

3. Response to comment: Figure files should contain only the image, as well as any associated keys/annotations. If legends are present within the figure files, please remove them. Figures should be provided as separate files, and each figure of a manuscript should be submitted as a single file.

Reply: We removed the images from the manuscript and submitted them separately.

Special thanks to you for your good comments.

Sincerely yours,

Zehua Lei, Dr
Professor and Chairman
Department of Hepatobiliary and Pancreatic Surgery
The Affiliated People's Hospital of Leshan City,
Leshan, Sichuan, 614000, PR China
e-mail: leitsehua@126.com
Tel: +86-13320921599